

https://doi.org/10.1093/pnasnexus/pgae067 Advance access publication 9 February 2024 Research Report

Even low levels of tree cover improve dietary quality in West Africa

Bowy den Braber (D^{a,*}, Charlotte Hall (D^{a,b}, Martin Brandt^a, Florian Reiner^a, Maurice Mugabowindekwe (D^a and Laura Vang Rasmussen (D^a

^aDepartment of Geosciences and Natural Resource Management, University of Copenhagen, Øster Voldgade 10, 1350 Copenhagen K, Denmark ^bDepartment of Biological and Environmental Sciences, University of Stirling, Stirling FK9 4LA, UK *To whom correspondence should be addressed: Email: bowydenbraber@gmail.com Edited By: Rebecca Seguin-Fowler

Abstract

Forests are attracting attention as a promising avenue to provide nutritious and "free" food without damaging the environment. Yet, we lack knowledge on the extent to which this holds in areas with sparse tree cover, such as in West Africa. This is largely due to the fact that existing methods are poorly designed to quantify tree cover in drylands. In this study, we estimate how various levels of tree cover across West Africa affect children's (aged 12–59 months) consumption of vitamin A–rich foods. We do so by combining detailed tree cover estimates based on PlanetScope imagery (3 m resolution) with Demographic Health Survey data from >15,000 households. We find that the probability of consuming vitamin A–rich foods increases from 0.45 to 0.53 with an increase in tree cover from the median value of 8.8 to 16.8% (which is the tree cover level at which the predicted probability of consuming vitamin A–rich foods at low levels of tree cover in the lowland forest-savanna ecoregions, whereas the difference between poor and non-poor is less pronounced in the Sahel-Sudan. These results highlight the importance of trees and forests in sustainable food system transformation, even in areas with sparse tree cover.

Significance Statement

Better quality diets are key to solving the widespread problem of nutrient deficiencies. Yet, far too often, food and nutrition security policies focus on increasing agricultural production and access to sufficient calories as the main solution, which often results in degraded ecosystems. In this study, we examine the empirical relationship between having forests and trees in the surroundings and the probability of children consuming nutritious foods. Based on a sample consisting of >15,000 rural households living in West Africa, we show that even low levels of tree cover improve the likelihood of children aged 12–59 months consuming vitamin A–rich foods.

Introduction

The current global food system is detrimental to both human and environmental health, with 800 million people undernourished and around 2 billion consuming low-quality diets that can lead to micronutrient deficiencies, and dramatically rising rates of noncommunicable diseases, such as diabetes (1, 2). At the same time, agriculture is responsible for about a third of global greenhouse gas emissions (3), is the leading driver of deforestation (4), is the largest user of global freshwater withdrawals, and is a major source of both freshwater and ocean pollution (5). A growing body of literature has shown the potential of forests and trees for providing dual benefits for both environmental sustainability and human health, primarily in middle- and low-income countries. For example, living in close proximity to forests has been shown to increase dietary diversity and consumption of nutrient-dense foods (6-10). Likewise, rural smallholders who incorporate trees on or around their farms have improved food security outcomes (11). However, there is a dearth of empirical evidence on whether these relationships hold in areas with sparse tree cover, despite such areas covering large parts of the world (12) and providing ecosystem services to large populations (13). Although multiple case studies from West Africa have indicated that even sparse tree cover could be positive for people's diets (14–18), methods to assess global tree cover (19) have, until now, been insufficient to quantify tree cover in drylands (12, 20), thus hindering large-scale assessments on tree-diet linkages. The recent development of methods in combination with access to imagery with a high enough resolution to identify single trees (12, 20, 21) provides an avenue to assess the effects of both forests and trees outside of forests on dietary quality. This is especially important in areas such as West Africa, our study area, because (i) there is a surprisingly high occurrence of trees outside of forests (20), albeit with their provision of a myriad of ecosystem services largely uncovered (22, 23), and (ii)

Competing Interest: The authors declare no competing interest.

Received: July 26, 2023. Accepted: January 29, 2024



© The Author(s) 2024. Published by Oxford University Press on behalf of National Academy of Sciences. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

rates of undernourishment and malnutrition are generally high in this region (2).

In this study, we examine the linkages between various levels of tree cover across West Africa and people's consumption of vitamin A-rich foods. We focus specifically on vitamin A-rich foods given the high rates of vitamin A deficiency across West Africa (24). Vitamin A deficiency can lead to a number of health issues, including permanent blindness, or, in less serious cases, "night blindness". West African countries (particularly Niger, Mali, and Guinea) had the highest rates of night blindness in pregnant women in 2005 (24) (unfortunately, there are no more recent statistics), with the exception of Ethiopia, which had the highest rate worldwide. We investigate whether the effects of tree cover on people's food consumption vary across ecoregions (Sahel-Sudan and lowland forest-savanna; Fig. S1) and people's poverty levels (measured as the living standards dimension of the multidimensional poverty index [MPI] (25)). We note that our data only comes from one single time point and that there is no experimental manipulation in our study design. As such, the effects of tree cover refer to associations in real-world settings, and we make no claims about causality. To examine the effects of tree cover on people's food consumption, we combine tree cover estimates from PlanetScope imagery (3 m resolution) with Demographic Health Survey (DHS) data from >80,000 households, of which >15,000 households have children aged 12-59 months, covering 10 countries in West Africa. Our approach uses a tree cover dataset based on state-of-the-art imagery (12), which allows us to also include areas with sparse tree cover, such as the Sahel. Because rainfall is an important determinant of tree cover in the region (26), we divide our study area into two regions with different rainfall patterns using ecoregions from Olson et al. (27). The Sahel-Sudan ecoregion in the north receives much less rainfall than the lowland forest-savanna ecoregion in the south and is characterized by different tree species (28). For both regions, we examine the linkages between tree cover and people's dietary quality across different poverty levels.

The underlying assumption for our analysis rests on recent research establishing that forests can provide considerable dietary benefits to households. That is, forests can improve people's diets along four key pathways (6, 29, 30). The first and most direct pathway is through the provision of wild foods, which include foods such as mangoes (Mangifera indica) and dark green leafy vegetables (e.g. baobab leaves [Adansonia digitata]) which tend to be very rich in vitamin A, and animal products (i.e. bushmeat and insects), which are a good source of bioavailable zinc and iron. Wild foods rarely make up the majority of the diet, instead supplementing what is available from agricultural production or markets. Wild foods thus tend to increase overall dietary diversity and can improve the micronutrient adequacy of the diet (31, 32). The second pathway is through income gains from sales of nontimber forest products (NTFPs), which can facilitate the purchase of nutritious foods from markets (33). The third pathway is through the flow of ecosystem services from forests into surrounding agricultural landscapes, which can in turn increase and/or diversify agricultural production (34). The fourth pathway is through the provision of fuelwood for cooking, which can improve nutrition by facilitating the preparation of a range of foods, particularly those with long cooking times (35), such as dark green leafy vegetables, often used in relish in West Africa. However, knowledge on the relative contribution of each of these four pathways remains limited, as existing studies have typically assessed these pathways in isolation and in areas with dense tree cover (6, 30). Due to the nature of the DHS data on which we rely, it is not possible to assess the

contribution of each of the four pathways. Yet, we advance existing knowledge by empirically assessing whether there is an association between tree cover and people's consumption of nutritious foods in areas with sparse tree cover. Such assessment is an important first step to uncover whether regions, such as the Sahel, with mostly sparse tree cover, exhibit the productivity and species diversity needed to improve diets.

We first examined the relationship between tree cover (in a 5-km radius around households) and the consumption of vitamin A-rich foods. The consumption of vitamin A-rich foods was a binary variable estimated using DHS data, whereby women were asked whether or not their children aged 12-59 months consumed vitamin A-rich foods within the last 24 h. We used a rigorous quasi-experimental matching technique (covariate balancing generalized propensity scores; CBGPS (36)) to adjust for the nonrandom distribution (selection bias) of tree cover, our treatment variable of interest. By controlling for key covariates likely to affect people's diets (Table S1), we were able to isolate the relationship between tree cover and the consumption of vitamin A-rich foods. The key advantage of the CBGPS approach is that it is applicable to continuous treatment variables, such as forest cover. We then combined this with generalized additive models (GAMs) (37), which have the advantage that they allow for nonlinear relationships. By combining these two approaches, our analysis presents an important step forward in teasing apart how various levels of tree cover affect consumption of a highly nutritious food group in nonlinear ways.

Results

Any increase in tree cover improves children's dietary quality

We found that higher levels of tree cover were associated with an increased probability of consuming vitamin A-rich foods (P < 0.001), especially in areas with very low levels of tree cover, where existing global assessments (19) have failed to accurately quantify tree cover (12). Specifically, our analysis revealed that an increase in tree cover from 8.8% (which is the median value across the >15,000 households with children aged 12-59 months) to 16.8% (the tree cover level at which the predicted probability of consuming vitamin A-rich foods is highest) is associated with an increase in the probability of consuming vitamin A-rich foods from 0.45 (credible interval 0.35-0.55) to 0.53 (credible interval 0.43-0.63; Fig. 1). This is a substantial increase given that just 41 and 50% of households consumed vitamin A-rich foods in the Sahel-Sudan and lowland forest-savanna ecoregions, respectively (Table S1). Moreover, we observed that when tree cover is above 30%, which is already classified as a forest (38), the effect of trees flattens. This suggests that in West Africa, trees outside of forests (i.e. low levels of tree cover that do not constitute a forest) play a more important role than trees inside forests for people's consumption of vitamin A-rich foods, which is also where most of the people reside. Our findings further advance existing studies showing that living near, and having access to, forest landscapes is beneficial for the intake of vitamin A-rich foods (17, 31, 39). Yet, our analysis shows that the effect is not linear, with the main effect occurring at sparse tree cover across West Africa.

The effects of tree cover vary across poverty levels and ecoregions

We also examined whether the relationship between tree cover and children's consumption of vitamin A-rich foods differed



Fig. 1. Effects of tree cover on consumption of vitamin A–rich foods. N = 15,875 rural households with children aged 12–59 months (81,296 rural households in total) in Senegal, Mali, Nigeria, Gambia, Benin, Guinea, Chad, Liberia, Sierra Leone, and Cameroon. The line indicates the fitted probability that the children in a household consumed vitamin A–rich foods. The shaded area shows the credible interval. The points indicate the average probability of consumption by clusters of households. The top chart shows the frequency of household observations across tree cover values, ranging from 39 to 4,579.

across two separate ecoregions (27) (the Sahel-Sudan ecoregion and the lowland forest-savanna ecoregion) and poverty levels. Overall, we see that in the Sahel-Sudan ecoregion, effects of tree cover are mostly equal among the poor and non-poor households, yet the non-poor benefit marginally more at 15-25% tree cover (P = 0.075, Fig. 2). Potential explanations for the non-poor benefiting more may include income gains from sales of NTFPs, which can facilitate the purchase of vitamin A-rich foods from markets (33) (the second pathway potentially connecting tree cover to people's diets). In contrast, in the lowland forest-savanna ecoregion, the poor are more likely to consume vitamin A-rich foods at low levels of tree cover (P < 0.001). While this observation is well aligned with research showing that the poorest people often rely on the forest for collecting and consuming wild foods (40) (the first pathway potentially connecting tree cover to people's diets), it adds to this literature by illustrating that it only happens at low levels of tree cover. That is, after the consumption of vitamin Arich foods peaks at ~18% tree cover, we observe a drop after which consumption levels remain relatively constant and the difference between poor and non-poor disappears. The fact that the non-poor households appear to be benefiting the most from sparse tree cover in the Sahel-Sudan ecoregion, whereas the opposite pattern is seen in the lowland forest-savanna ecoregion, can be explained by different characteristics of sparse tree cover. That is, in the lowland forest-savanna ecoregions, low tree cover levels likely represent deforested areas that host semi-agricultural species such as vitamin A-rich dark green leafy vegetables (41) that can be collected for free (by the poor). Such species are generally less common in areas with naturally low tree cover, such as the Sahel-Sudan ecoregion. Low levels of tree cover in the lowland forest-savanna ecoregion could also represent agroforestry systems that host trees yielding vitamin Arich fruits such as mango.

Discussion

Our analysis reveals the importance of trees in areas with sparse tree cover, yet these are rarely taken into account in existing forest-diet research, primarily due to limitations in remote sensing methods that are often not designed to quantify tree cover in drylands (12). While we recognize the potential dietary benefits of trees outside of forests in landscapes with relatively sparse tree cover, it is also critical to acknowledge that existing large-scale environmental policies in Africa, such as the AFR100 and the Great Green Wall restoration initiatives, are biased toward financing reforestation initiatives focused on the planting of highly visible, fast growing tree species (42) instead of food-trees. Future initiatives could be better targeted to preserve and increase species with nutritional benefits (43). In the Sahel-Sudan ecoregions, examples include Moringa oleifera and A. digitata (baobabs)—both species that are able to grow in areas with sparse tree cover. Other examples of commonly observed species include Annona



Fig. 2. Effects of tree cover on consumption of vitamin A-rich foods per ecoregion and across poverty levels. N = 15,875 rural households with children under 5 years old (81,296 rural households in total) in Senegal, Mali, Nigeria, Gambia, Benin, Guinea, Chad, Liberia, Sierra Leone, and Cameroon. Lines indicate the fitted probability that the children in a household consumed vitamin A-rich foods—red line indicates a high level of poverty (MPI > 0.83— highest 20% of the population), blue line indicates a low level of poverty (MPI < 0.33—lowest 20% of the population). The shaded area shows the credible interval. The points indicate the average probability of consumption of groups of households with similar tree cover levels and MPI. The top charts show the frequency of household observations across tree cover values, ranging from 1 to 545 for the Sahel-Sudan ecoregions (N = 5,965), and from 11 to 136 for the Lowland Forest-Savanna ecoregions (N = 9,910).

senegalensis (wild custard apple) and Salvadora persica (the toothbrush tree) (Table S2). Trees in the Sahel-Sudan might be especially important for other purposes too because they also serve as an important source of fodder, fuelwood, and fiber, and provide regulating and supporting ecosystem services (22). In lowland forestsavanna areas with sparse tree cover, reforestation initiatives will have to account for the fact that these areas have likely already been deforested and any species of value will have been removed. Instead, such areas are likely inhabited by earlysuccessional pioneer species, and invasive and exotic species (many of which are edible). As such, our results highlight how different ecoregions in West Africa exhibit different patterns in the benefits people can obtain from trees.

Another simple, yet far-reaching result, is that increases in tree cover above 30% will not necessarily increase children's consumption of vitamin A-rich foods. While we acknowledge that higher tree cover is critical for biodiversity, our results challenge the assumption of a linear relationship between tree cover and people's dietary diversity (39). This result is similar to the findings from Ickowitz et al. (9) who showed how consumption of dark green leaves peaks at relatively low levels of tree cover before declining. Our results also support previous assessments showing that areas outside forests, such as savanna areas, may provide important ecosystem services likely to be beneficial to diets (13). Yet, savannas are often neglected in restoration efforts as they are often seen as degraded forests (28, 44). The finding that high tree cover might do little to secure the consumption of vitamin A-rich foods can be due to a number of possible explanations. First, forest-based pollinators increase the production of domestic fruits in nearby areas, meaning that many scattered trees around the landscape might lead to more effective pollination than bigger areas of forest (45). Second, households are more prone to collect wild fruits and vegetables when traversing landscapes with many scattered trees rather than going deep into interior forest (40). This has, for example, been observed in Nigeria, where a recent study showed how vitamin A-rich foods (e.g. bush mango [Irvingiaceae]) were sourced more frequently among forest-edge communities when compared with interior forest communities (18). Third, in landscapes with sparse tree cover, there might be smaller areas of "managed forests" where valuable fruit trees are consciously maintained (46). Fourth, landscapes with very sparse tree cover could fail to supply diverse vitamin A-rich foods, resulting in people purchasing more of these foods. Finally, landscapes with high tree cover might include forests that are managed for conservation with restricted access for communities (47). Future work should seek to create more explicit evidence on which of these explanations are prevalent in the Sahel-Sudan and lowland forest-savanna ecoregions.

Moreover, our analysis sheds light on the effects that trees can have on dietary quality across poverty levels. In the Sahel-Sudan, we show that increased tree cover has a positive effect across all poverty levels. This suggests that the absence of access to (free) vitamin A-rich foods from trees might not always be compensated by purchasing such foods instead. It also underlines the importance of trees for the intake of vitamin A for a large population in the Sahel-Sudan. Nevertheless, we observe that although higher tree cover in general is beneficial, the least poor households appear to benefit marginally more from greater tree cover than the

poorest. Tree-planting and regrowth initiatives, such as the Great Green Wall, should therefore ensure that additional trees would benefit all layers of society (42). Such focus is timely given the shift in species composition recently observed in the Sahel-Sudan; shrubs and highly drought-tolerant and exotic tree species are increasing, whereas traditionally used, multifunctional species and larger trees are decreasing (48). In the lowland forest-savanna ecoregion, we see that the poorest people benefit the most from relatively low levels of tree cover, while the least poor benefit from higher levels of tree cover. In contrast to the Sahel-Sudan, low levels of tree cover in the forest-savanna ecoregion might still host dark green leaves, which are an important and easily accessible source of vitamin A, especially for the poor (Fig. S2). Any treeplanting and regrowth initiatives in the lowland forest-savanna, especially in deforested areas, would have to ensure that benefits from additional tree cover also spread to the poorest people.

In summary, our analysis links data on trees inside and outside of forests to people's food consumption and poverty levels on a broad scale. We show how the presence of trees can potentially be important for the intake of vitamin A-rich foods-but that the effect is not linear as it levels off with more tree cover in the landscape. The next research step could be to examine how the presence of trees influences agricultural diversity, yields, and crop nutritional content (49). New remote sensing techniques that identify specific tree species would allow us to untangle the effects of trees and identify trees with the highest contribution to dietary quality (50). Moreover, studies linking trees outside of forests to people's diets would benefit from more detailed food consumption data recording the quantity of different foods consumed. Such data would allow for a better understanding of how trees affect people's micronutrient intake-not just their intake of specific foods. This is particularly important in West Africa, where micronutrient deficiencies are widespread, with 57% of the population being moderate or severe food insecure and 86% of the population unable to afford a healthy diet (2). Lastly, future studies would benefit from gathering data on where households source their food from, to illuminate whether (and to what extent) people are (i) consuming nutritious foods directly from trees scattered in the landscape or more dense forests or (ii) purchasing more nutritious foods.

Methods

Data

We used DHS household data from 10 countries in western Africa (Table S3). The DHS provides information on a range of socioeconomic and health variables and is available at https:// dhsprogram.com/data/available-datasets.cfm. In total, this study uses DHS data from 15,875 households from 2,587 rural clusters with at least one child aged 12–59 months. Clusters roughly correspond to a village. DHS data were collected between 2014 and 2019 across the 10 countries in western Africa. Geolocations are available for all clusters, but are displaced for confidentiality purposes. The displacement in a random direction is up to 5 km for 99% of the clusters, and up to 10 km for the remaining 1% of the clusters. To account for the displacement in geolocations, we used a 5-km buffer around each geolocation when spatially overlaying the household data with tree cover data and other spatial data.

Consumption of vitamin A-rich foods

The DHS interviewed women in the households about their children's food consumption over the previous 24 h. Women were asked whether their children had consumed any foods from 18 predetermined food groups (e.g. eggs and dairy). We chose to focus on the consumption of vitamin A-rich foods by children aged 12–59 months in this study due to (i) forests (and trees outside of forests) anticipated to host a multitude of vitamin A-rich foods, such as wild mango, and (ii) the high rates of vitamin A deficiency in many parts of Africa, which makes it paramount to identify avenues to improve vitamin A intake (24). Thus, we combined the three DHS food groups—vitamin A-rich fruits, vitamin Arich vegetables and tubers—and dark green leafy vegetables to assess children's overall intake of vitamin A-rich foods. For the analysis presented in Fig. S2, we only included the consumption of dark green leafy vegetables.

Tree cover

We used tree cover data from a recent continental map of African tree cover in 2019 (12). Based on a deep-learning model applied to 3 m PlanetScope imagery, this dataset includes tree cover from both forest and non-forest trees. For our analysis at the household level, the binary tree cover data were first aggregated to percent cover at 100 m resolution. The tree cover variable was then calculated as the mean percent tree cover in a 5-km buffer around each household.

Living standards

We were specifically interested in whether the effect of tree cover on the consumption of vitamin A–rich foods would vary across different levels of poverty. We used the living standards dimension of the MPI (MPI-LS (25)), as an indicator of the asset poverty of each household (Table S4). The living standards dimension varies from 0 to 1, with 1 being the most deprived. The MPI-LS is composed of six indicators: (i) assets, (ii) electricity, (iii) sanitation, (iv) cooking fuel, (v) water source, and (vi) housing. Each indicator has an equal weight.

Ecoregion

We only included ecoregion as a covariate in the model that included an interaction with ecoregion (presented in Fig. 2) and did not differentiate between ecoregions in our main analysis (presented in Fig. 1). Our study area covers four major ecoregions (27), which we lumped into two main ecoregions that represent the ecosystems in the north and south of our study area. Our first region consists of the Sahelian acacia savanna and Sudanian savanna (west and east) ecoregions. Our second region consists of the forest-savanna (Guinean and north-Congolian) and lowland forest (western Guinean, eastern Guinean, and Nigerian) ecoregions. We removed 319 clusters that overlapped ecoregions that are very small (e.g. deserts and xeric shrublands, montane forests, flooded savannas, and mangroves). We calculated the overlap of each cluster (and thus household) with each main ecoregion and assigned each household to the ecoregion with the largest overlap. To increase the contrast in this analysis, we removed 127 clusters that overlapped our two main ecoregions.

Additional covariates

We controlled for a suite of biophysical and socioeconomic covariates known to influence people's diets—and thus potentially their consumption of vitamin A–rich foods. First, we controlled for household characteristics that are important predictors of household food consumption: household size (31), age of the household head (51), number of children under 5 years old, and the education dimension of the MPI (52) (Table S4). Second, we controlled for agricultural characteristics (53): from the DHS, we extracted tropical livestock units (54) and agricultural land cover from Zanaga et al. (55). Third, we controlled for whether a conflict occurred in the past year before the survey, as (i) conflicts have occurred in many West African countries over the last decade (56), and (ii) conflicts are known to influence people's food intake (57). Fourth, we included geographical covariates that influence land suitability and access (58): slope, percentage of area covered by water, population density, and travel time to the nearest densely populated area. Geographical variables were calculated as the mean value within 5 km of the geolocated household. Finally, we included the country and season (dry/wet) during which the survey was conducted as dummy variables. Additional descriptions and data sources for all covariates are available in Table S5.

Analysis

We used matching and nonlinear regression techniques to assess the effects of tree cover on the consumption of vitamin A–rich foods and dark green leafy vegetables. We conducted all statistical analyses in R (59).

We used CBGPS (36) to generate weights that adjust for the nonrandom distribution of tree cover, our predictor variable of interest. CBGPS is a form of propensity score matching and, in contrast to most other forms of matching, can be used when the predictor variable of interest is continuous. The CBGPS method produces weights that minimize the correlation between the treatment (tree cover) and covariates (Fig. S3). These weights are then used in subsequent models to reduce endogeneity between tree cover and other covariates, thereby reducing potential bias in the estimates of tree cover on consumption of vitamin A-rich foods.

After matching, we used GAMs to analyze the effect of tree cover on the dependent variable. GAMs have the advantage that they allow for nonlinear relationships. Linear models use a single coefficient to model the relationship between predictor and outcome variables, but GAMs use a smooth function that allows the relationship to vary along the gradient of the predictor. Therefore, by using GAMs, we can assess the effect of tree cover across the entire gradient from bare ground to forest at each percentage of tree cover. We used the mgcv package in R to fit GAMs (37). In our main model setup, we included parametric terms for country, education, season, and conflict. We then included numeric covariates as cubic regression smoothing splines, because they take into account both smoothness and local influence, and are widely used in models for nonlinear data (60). To reduce spatial autocorrelation, we included geographic coordinates (latitude and longitude) as a (default) thin smoothing spline.

Our main model takes the form:

 $g(\mu_i) = X_i\beta + f_m(T_m) + f_1(x_1) + \dots + f_m(x_m) + f(\text{lat, long}),$

where $g(\mu_i)$ represents the response variable, $X_i\beta$ is the parametric part of the linear predictor, $f_m(T_m)$ represents the modeled smooth of tree cover T_m , $f_m(x_m)$ are modeled smooths of numeric covariates, and f(lat, long) is the modeled smooth of geographic location.

We used the bam function for large datasets, the default fREML method, the argument discrete = TRUE to ensure model convergence, and a quasi-binomial distribution to model the consumption of vitamin A-rich foods.

Because GAMs do not assume linear relationships, they are also better at capturing complicated interactions between predictor variables than linear models. We assessed the interaction between tree cover and living standards by adding an interaction effect. We did so by adding a tensor product smooth ("ti" function) to the main model setup. To assess the interaction effect separately for each ecoregion, we also included ecoregion as a parametric term and allowed the interaction (i.e. tensor product smooth) to vary for each ecoregion. Our model thus takes the form:

$$g(\mu_i) = X_i\beta + f_m(T_m) + f_1(x_1) + \dots + f_m(x_m) + f_{geo}(\text{lat, long})$$

+ $f_m(T_m, v_m)Z_m,$

where $f_m(T_m, v_m)$ represents the interaction between tree cover T_m and living standards v_m , and Z_m is the ecoregion.

Finally, we used the DHARMa package (61) to calculate Moran's *I* as an estimate of spatial autocorrelation. Although spatial autocorrelation was still present in the residuals after including a smooth of the geographic location, Moran's *I* was very low (0.01). We thus conclude that there is no evidence that spatial autocorrelation influences our estimates.

Supplementary Material

Supplementary material is available at PNAS Nexus online.

Funding

L.V.R., B.d.B., and C.H. were funded by the European Research Council under the European Union's Horizon 2020 Research and Innovation Programme (grant agreement 853222 FORESTDIET). M.B., M.M., and F.R. were supported by the European Research Council under the European Union's Horizon 2020 Research and Innovation Programme (grant agreement no. 947757 TOFDRY) and a DFF Sapere Aude grant (no. 9064–00049B).

Author Contributions

The objectives and methodology for this analysis were conceived and designed by B.d.B., C.H., and L.V.R. B.d.B. conducted the analyses with support from C.H. and L.V.R. M.B., F.R., and M.M. contributed data on tree cover. B.d.B., C.H., and L.V.R. wrote the manuscript with help from all authors. All authors contributed to editing subsequent drafts.

Data Availability

All data needed to replicate our results are available online. Our analysis uses publicly available data from the Demographic Health Survey (DHS), available at https://dhsprogram.com/data/ available-datasets.cfm. Maps on tree cover are available at https://doi.org/10.5281/zenodo.7764460. Ecoregion data are available from https://ecoregions.appspot.com/. Geographic data on covariates were downloaded using Google Earth Engine (see Table S5 for references). Code is available from B.d.B. (bowydenbraber@gmail.com).

References

- GBD 2017 Diet Collaborators. 2019. Health effects of dietary risks in 195 countries, 1990–2017: a systematic analysis for the Global Burden of Disease Study 2017. *Lancet.* 393:1958–1972.
- 2 FAO, IFAD, UNICEF, WFP, WHO. 2022. The state of food security and nutrition in the world 2022: repurposing food and agricultural policies to make healthy diets more affordable. Rome: FAO, IFAD, UNICEF, WFP, WHO. https://doi.org/10.4060/cc0639en
- 3 Crippa M, et al. 2021. Food systems are responsible for a third of global anthropogenic GHG emissions. Nat Food. 2:198–209.

- 4 Pendrill F, et al. 2022. Disentangling the numbers behind agriculture-driven tropical deforestation. *Science*. 377:eabm9267.
- 5 Campbell BM, *et al.* 2017. Agriculture production as a major driver of the Earth system exceeding planetary boundaries. *E&S.* 22: art8.
- 6 Gergel SE, et al. 2020. Conceptual links between landscape diversity and diet diversity: a roadmap for transdisciplinary research. *Bioscience*. 70:563–575.
- 7 Hall CM, et al. 2022. Deforestation reduces fruit and vegetable consumption in rural Tanzania. Proc Natl Acad Sci U S A. 119: e2112063119.
- 8 Rasmussen LV, et al. 2020. Forest pattern, not just amount, influences dietary quality in five African countries. *Glob Food Sec.* 25: 100331.
- 9 Ickowitz A, Powell B, Salim MA, Sunderland TCH. 2014. Dietary quality and tree cover in Africa. *Glob Environ Change*. 24:287–294.
- 10 Cheek JZ, et al. 2023. Wild foods contribute to women's higher dietary diversity in India. Nat Food. 4:476–482.
- 11 Vansant EC, et al. 2022. What are the links between tree-based farming and dietary quality for rural households? A review of emerging evidence in low- and middle-income countries. *People* Nat. 4:296–311.
- 12 Reiner F, et al. 2023. More than one quarter of Africa's tree cover is found outside areas previously classified as forest. Nat Commun. 14:2258. [dataset]
- 13 Malmborg K, Sinare H, Enfors Kautsky E, Ouedraogo I, Gordon LJ. 2018. Mapping regional livelihood benefits from local ecosystem services assessments in rural Sahel. PLoS One. 13:e0192019.
- 14 Binam JN, et al. 2015. Effects of farmer managed natural regeneration on livelihoods in semi-arid West Africa. Environ Econ Policy Stud. 17:543–575.
- 15 Kalinganire A, Weber JC, Uwamariya A, Kone B. 2008. Improving rural livelihoods through domestication of indigenous fruit trees in the Parklands of the Sahel. In: Akinnifesi F, et al., editors. Indigenous fruit trees in the tropics: domestication, utillization and commercialization. Wallingford: CABI Publishing. p. 186–203.
- 16 Sop TK, Oldeland J, Bognounou F, Schmiedel U, Thiombiano A. 2012. Ethnobotanical knowledge and valuation of woody plants species: a comparative analysis of three ethnic groups from the sub-Sahel of Burkina Faso. Environ Dev Sustain. 14:627–649.
- 17 Fungo R, et al. 2016. Contribution of forest foods to dietary intake and their association with household food insecurity: a crosssectional study in women from rural Cameroon. Public Health Nutr. 19:3185–3196.
- 18 Friant S, et al. 2019. Life on the rainforest edge: food security in the agricultural-forest frontier of cross river state, Nigeria. Front Sustain Food Syst. 3:113.
- 19 Hansen MC, et al. 2013. High-resolution global maps of 21st-century forest cover change. *Science*. 342:850–853.
- 20 Brandt M, et al. 2020. An unexpectedly large count of trees in the West African Sahara and Sahel. *Nature*. 587:78–82.
- 21 Li S, et al. 2023. Deep learning enables image-based tree counting, crown segmentation, and height prediction at national scale. PNAS Nexus. 2:pgad076.
- 22 Mbow C, Halle M, El Fadel R, Thiaw I. 2021. Land resources opportunities for a growing prosperity in the sahel. *Curr Opin Environ Sustain*. 48:85–92.
- 23 Skole DL, Mbow C, Mugabowindekwe M, Brandt MS, Samek JH. 2021. Trees outside of forests as natural climate solutions. Nat Clim Chang. 11:1013–1016.
- 24 WHO. 2009. Global prevalence of vitamin A deficiency in populations at risk 1995–2005: WHO global database on vitamin A deficiency. World Health Organization.

- 25 Alkire S, Foster J. 2011. Counting and multidimensional poverty measurement. J Public Econ. 95:476–487.
- 26 Sankaran M, et al. 2005. Determinants of woody cover in African savannas. Nature. 438:846–849.
- 27 Olson DM, et al. 2001. Terrestrial ecoregions of the world: a new map of life on EarthA new global map of terrestrial ecoregions provides an innovative tool for conserving biodiversity. BioScience. 51:933–938.
- 28 Aleman JC, et al. 2020. Floristic evidence for alternative biome states in tropical Africa. Proc Natl Acad Sci U S A. 117:28183–28190.
- 29 Olesen RS, Hall CM, Rasmussen LV. 2022. Forests support people's food and nutrition security through multiple pathways in low- and middle-income countries. One Earth. 5:1342–1353.
- 30 Baudron F, et al. 2019. Testing the various pathways linking forest cover to dietary diversity in tropical landscapes. Front Sustain Food Syst. 3:97.
- 31 Powell B, Maundu P, Kuhnlein HV, Johns T. 2013. Wild foods from farm and forest in the east usambara mountains, Tanzania. Ecol Food Nutr. 52:451–478.
- 32 Vinceti B, Eyzaguirre P, Johns T. 2008. The nutritional role of forest plant foods for rural communities. In: Colfer C, editor. *Human health and forests*. London: Earthscan. p. 63–96.
- 33 Nandi R, Nedumaran S, Ravula P. 2021. The interplay between food market access and farm household dietary diversity in low and middle income countries: a systematic review of literature. *Glob Food Sec.* 28:100484.
- 34 Reed J, et al. 2017. Trees for life: the ecosystem service contribution of trees to food production and livelihoods in the tropics. For Policy Econ. 84:62–71.
- 35 Waswa F, Mcharo M, Mworia M. 2020. Declining wood fuel and implications for household cooking and diets in Tigania subcounty Kenya. Sci Afr. 8:e00417.
- 36 Fong C, Hazlett C, Imai K. 2018. Covariate balancing propensity score for a continuous treatment: application to the efficacy of political advertisements. Ann Appl Stat. 12:156–177.
- 37 Wood S. 2012. mgcv: Mixed GAM Computation Vehicle with GCV/ AIC/REML smoothness estimation. http://cran.r-project.org/ web/packages/mgcv/index.html.
- 38 FAO. 2022. The State of the World's Forests 2022: forest pathways for green recovery and building inclusive, resilient and sustainable economies. Rome: FAO. https://doi.org/10.4060/cb9360en.
- 39 Hall C, et al. 2019. The relationship between forest cover and diet quality: a case study of rural southern Malawi. Food Sec. 11: 635–650.
- 40 Hickey GM, Pouliot M, Smith-Hall C, Wunder S, Nielsen MR. 2016. Quantifying the economic contribution of wild food harvests to rural livelihoods: a global-comparative analysis. *Food Policy*. 62: 122–132.
- 41 Maseko H, Shackleton CM, Nagoli J, Pullanikkatil D. 2017. Children and wild foods in the context of deforestation in rural Malawi. Hum Ecol Interdiscip J. 45:795–807.
- 42 Löfqvist S, et al. 2023. How social considerations improve the equity and effectiveness of ecosystem restoration. *BioScience*. 73:134–148.
- 43 Sacande M, Berrahmouni N. 2016. Community participation and ecological criteria for selecting species and restoring natural capital with native species in the Sahel. Restor Ecol. 24:479–488.
- 44 Bond WJ, Stevens N, Midgley GF, Lehmann CER. 2019. The trouble with trees: afforestation plans for Africa. Trends Ecol Evol. 34: 963–965.
- 45 Garibaldi LA, *et al.* 2013. Wild pollinators enhance fruit set of crops regardless of honey bee abundance. *Science*. 339:1608–1611.

- 46 Padoch C, Pinedo-Vasquez M. 1996. Smallholder forest management: looking beyond non-timber forest products. In: Ruiz Perez M, Arnold J, editors. Current issues in non-timber forest products research. Bogor: CIFOR. p. 103–118.
- 47 Ickowitz A, et al. 2022. Transforming food systems with trees and forests. Lancet Planet Health. 6:e632–e639.
- 48 Goffner D, Sinare H, Gordon LJ. 2019. The Great Green Wall for the Sahara and the Sahel Initiative as an opportunity to enhance resilience in Sahelian landscapes and livelihoods. *Reg Environ Change*. 19:1417–1428.
- 49 Wood SA, Tirfessa D, Baudron F. 2018. Soil organic matter underlies crop nutritional quality and productivity in smallholder agriculture. Agric Ecosyst Environ. 266:100–108.
- 50 Gamfeldt L, et al. 2013. Higher levels of multiple ecosystem services are found in forests with more tree species. Nat Commun. 4: 1340.
- 51 Mango N, Zamasiya B, Makate C, Nyikahadzoi K, Siziba S. 2014. Factors influencing household food security among smallholder farmers in the Mudzi district of Zimbabwe. *Dev South Afr.* 31: 625–640.
- 52 Torheim LE, et al. 2004. Nutrient adequacy and dietary diversity in rural Mali: association and determinants. *Eur J Clin Nutr.* 58: 594–604.
- 53 Jones AD, Shrinivas A, Bezner-Kerr R. 2014. Farm production diversity is associated with greater household dietary diversity in

Malawi: findings from nationally representative data. Food Policy. 46:1-12.

- 54 FAO. 2018. Guidelines on methods for estimating livestock production and productivity. Rome: FAO.
- 55 Zanaga D, *et al*. 2022. ESA WorldCover 10 m 2021 v200. Zenodo [dataset]. https://doi.org/10.5281/zenodo.7254221.
- 56 Gleditsch NP, Wallensteen P, Eriksson M, Sollenberg M, Strand H. 2002. Armed conflict 1946–2001: a new dataset. J Peace Res. 39: 615–637. [dataset]
- 57 Martin-Shields CP, Stojetz W. 2019. Food security and conflict: empirical challenges and future opportunities for research and policy making on food security and conflict. World Dev. 119: 150–164.
- 58 Angelsen A. 2010. Policies for reduced deforestation and their impact on agricultural production. Proc Natl Acad Sci U S A. 107: 19639–19644.
- 59 R. C. Team. 2018. R: a language and environment for statistical computing. Vienna (Austria): R Foundation for Statistical Computing. 2012. http://www.R-project.org.
- 60 Grajeda LM, et al. 2016. Modelling subject-specific childhood growth using linear mixed-effect models with cubic regression splines. *Emerg Themes Epidemiol*. 13:1.
- 61 Hartig F, Hartig MF. 2017. DHARMa: residual diagnostics for hierarchical (multi-level/mixed) regression models. https://cran.rhub.io/web/packages/DHARMa/index.html.